Author's response to reviews

Title: Reconstruction of the coracoacromial ligament during a modified Latarjet procedure

Authors:

Matthias E Aurich (kontakt@matthias-aurich.de)
Gunther O Hofmann (gunther.hofmann@med.uni-jena.de)
Florian Gras (florian.gras@med.uni-jena.de)

Version: 2 Date: 7 April 2015

Author's response to reviews: see over
MS: 2881742351668805

Revised manuscript entitled: "Reconstruction of the coracoacromial ligament in a modified Latarjet procedure. Surgical technique and early results."

Dear Editor,

please find attached my revised manuscript entitled: "Reconstruction of the coracoacromial ligament in a modified Latarjet procedure. Surgical technique and early results."

I have made the following revisions according to your Email dated April 7, 2015.

Title page:
- The Emails of all authors have been included (line 20, 26, 31)
- Authors contributions have been included (line 35 to 39)
- Acknowledgements have been included (line 41, 42)

Ethics approval:
- The information on ethics approval is included in the methods section of the manuscript (line 114 and 115)

I hope that I have made the changes correctly.

With kind regards,
Matthias E. Aurich, MD, DSc